Author's response to reviews

Title: A retrospective study of central nervous system shunt infections diagnosed in a university hospital during a 4-year period

Authors:

Suzan Sacar (suzansacar@yahoo.com)
Huseyin Turgut (huturgut@yahoo.com)
Semra Toprak (drsemratoprak@hotmail.com)
Bayram Cirak (bayramcirak@yahoo.com)
Erdal Coskun (ercoskun@yahoo.com)
Ozlem Yilmaz (ozlem-23@mynet.com)
Koray Tekin (koraytekin@hotmail.com)

Version: 10 Date: 8 March 2006

Author's response to reviews: see over
Dear Editor,

We would like to submit the final revised manuscript entitled “A RETROSPECTIVE STUDY OF CENTRAL NERVOUS SYSTEM SHUNT INFECTIONS DIAGNOSED IN A UNIVERSITY HOSPITAL DURING A 4-YEAR PERIOD” for possible publication in BMC Infectious Diseases.

In the 12. reference was checked again and the missing names of the authors (Palabiyikoglu I, Tekeli E, Cokca F, Akan O, Unal N) were added. The right version is:


Thank you again for your kind support.

With best wishes:

Suzan Sacar, MD

Pamukkale University, Department of Infectious Diseases and Clinical Microbiology, Denizli, Turkey